Author’s response to reviews

Title: Prognostic factors for survival in patients with malignant pleural effusions. What is considered important for treatment?

Authors:

Mauro M Zamboni (mauro.zamboni@gmail.com)
Cyro T Silva Jr (ctsilvajunior@predialnet.com.br)
Rodrigo Baretta (rodrigobaretta@hotmail.com)
Edson T Cunha (edsontoscano@gmail.com)
Gilberto P Cardoso (gilpcard@globo.com)

Version: 2 
Date: 19 September 2014

Author’s response to reviews:

Instituto Nacional de Câncer
Rua Sorocaba 464/302
CEP 22271-110
Rio de Janeiro – Brazil
mauro.zamboni@gmail.com

Catia Cornacchia
Executive Editor
BMC Pulmonary Medicine

September 16, 2014

Dear Dr. Cornacchia:

I am pleased to submit an original research article entitled “Prognostic factors for survival in patients with malignant pleural effusions. What is considered important for treatment?” by Cyro T. S. Junior; Rodrigo Baretta; Edson T.Cunha; Gilberto P. Cardoso and Mauro M. Zamboni for consideration for publication in the BMC Pulmonary Medicine.

As requested by Journal Editorial Office all the changes were been done: Title Page; Abstract Section; Article Section; Competing Interests; Author’s Contributions; Acknowledgements and Line Numbering.

Thank you for your consideration!

Sincerely,